Author’s response to reviews

Title: Menstrual Health and Hygiene among Indigenous Australian girls and women: Barriers and opportunities

Authors:

Emily Krusz (e.krusz@uq.edu.au)

Nina Hall (n.hall2@uq.edu.au)

Dani Barrington (d.j.barrington@leeds.ac.uk)

Sandra Creamer (s.creamer@uq.edu.au)

Wendy Anders (seniorprojects@natsiwa.org.au)

Minnie King (womenoncountry@bigpond.com)

Helen Martin (womenoncountry1@bigpond.com)

Julie Hennegan (jheneg1@jhu.edu)

Version: 3 Date: 29 Sep 2019

Author’s response to reviews:

Thank you for the editorial guidance. In response, we have:

- replaced 'Introduction' with 'Background';
- ensured that the email addresses of Sandra Creamer and Helen Martin
- provided a 'clean copy'.